Author’s response to reviews

Title: Impact of clinical trial participation on survival in patients with castration-resistant prostate cancer: A multi-center analysis

Authors:

Kyo Chul Koo (gckoo@yuhs.ac)
Jong Soo Lee (js1129@yuhs.ac)
Jong Won Kim (doctor2play@yuhs.ac)
Kyung Suk Han (urokoo@gmail.com)
Kwang Suk Lee (calmenow@yuhs.ac)
Do Kyung Kim (dokyung@yuhs.ac)
Yoon Soo Hah (urohah@yuhs.ac)
Koon Ho Rha (khrha@yuhs.ac)
Sung Joon Hong (sjhong346@yuhs.ac)
Byung Ha Chung (chung646@yuhs.ac)

Version: 3 Date: 13 Apr 2018

Author’s response to reviews:

Editor Comments

1. Title Page:

Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

REPLY: The E-mail addresses of all authors have been described in the title page.

2. Abstract:

We note that your abstract has been published in the European Urology Supplements (https://www.sciencedirect.com/science/article/pii/S1569905618315860?via%3Dihub). As per our editorial policies, please cite your previously published abstract.
REPLY: The above abstract has been cited in the Discussion section (Discussion section, page 8, line 34)

3. Headings:

a. Please change ‘Introduction’ to ‘Background’
REPLY: ‘Introduction’ has been changed to ‘Background’.

b. Please change ‘Materials and Methods’ to ‘Methods’
REPLY: ‘Materials and Methods’ has been changed to ‘Methods’.

4. Ethics Approval and Consent to Participate:

In the Declarations, please also include the full name of the ethics committee that waived the need for consent.

REPLY: The full name of the ethics committee which waived the need for consent has been described in the Declarations section (Declaration section, page 10, lines 28 and 29) as; “Informed consent was waived from the Yonsei University Health System Institutional Review Board since patients’ information was collected during the routine clinical practice and patients were identified by anonymized investigator-generated code not linkable to their personal data.”

5. Permission to access databases:

In the “Ethics Approval and Consent to Participate”, please indicate who granted you access to the institutional databases and the other databases used in the study. Were these the appropriate authorities?

REPLY: The Yonsei University Health System Institutional Review Board granted us to access to the institutional databases and the other databases used in the study. This information was added to the Ethics approval and consent to participate section (page 11, lines 1 and 2). This Institutional Review Board had the appropriate authority for granting access to all institutional databases.

6. Funding:

In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
REPLY: We have added to the Funding section (page 10, lines 12-14), that the funding body had no active role in any stage of the study, including: design, data collection, analysis, interpretation of data, and writing the manuscript.

7. Authors’ Contributions:

a. We note that JW Kim is missing from this section. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship.

REPLY: This author (JW Kim) has been added to the Author’s contribution section (page 10, line 19).

b. Please note that based on the current authors’ contributions section, several listed authors do not automatically qualify for authorship.

We ask that you ensure that author contributions are in line with the ICMJE guidelines (below), and that all listed authors have performed all four points specified below.

REPLY: We note that not all authors automatically qualified for authorship according to the information that we had provided (intended for brevity of the manuscript). We have provided full information on each author’s contribution to the Authors’ contributions section (page 10).

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. According to the ICMJE guidelines, to qualify as an author one should have:

a) made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; AND

b) been involved in drafting the manuscript or revising it critically for important intellectual content; AND

c) given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content;

d) agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Anyone listed as an author must be included in this section. If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship.
Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section.

8. File Inventory:

Thank you for providing a cover letter and document containing your responses to the reviewers’ comments. These are no longer required, so please remove them from the File Inventory.

REPLY: The corresponding documents have been removed from the File Inventory.